

# Diabetic foot problems: prevention and management

Information for the public Published: 26 August 2015 Last updated: 18 January 2022

www.nice.org.uk

# Diabetic foot problems: the care you should expect

Type 1 and type 2 diabetes can cause problems with blood flow and damage your nerves. This can cause pain, tingling or numbness in your feet. If you lose feeling in your feet, you may not notice if you hurt them or if you get sores or ulcers. Any ulcers you get will take longer to heal. You may find it difficult to walk, and in extreme cases, your foot may be so badly damaged that it has to be removed.

We want this guideline to make a difference for children, young people and adults with type 1 or type 2 diabetes by making sure:

- they get annual foot checks from age 12 onwards
- people who are at higher risk get seen quickly by the foot protection service, for more detailed advice and support

- they get the best possible treatments for foot ulcers and foot infections
- people with suspected Charcot arthropathy (a serious condition that weakens the foot bones, leading to dislocations, fractures and changes in the shape of the foot) are seen by specialists within 2 working days.

### Making decisions together

Decisions about treatment and care are best when they are made together. Your healthcare professionals should give you clear information, talk with you about your options and listen carefully to your views and concerns.

To help you make decisions, think about:

- What matters most to you what do you want to get out of any assessments or treatment?
- What are you most worried about are there risks or downsides to the treatment that worry you more than others?
- How will the treatment affect your day-to-day life?
- What happens if you do not want to have treatment?

If you need more support to understand the information you are given, tell your healthcare professional.

Read more about making decisions about your care.

### Your diabetes care services

The foot protection service and the multidisciplinary foot care service both provide specialist foot care for people with diabetes.

The doctors, nurses and other healthcare professionals who provide diabetes care should take into account if you have any disabilities, for example, if you have any sight problems. In addition, if you are housebound or living in a care home or nursing home, they should take this into account to make sure you have the same foot checks and treatments as everyone else.

### Foot protection service

The foot protection service is usually based outside the hospital, for example, in a health centre or GP clinic. The foot protection service specialises in providing foot care for people with diabetes, preventing diabetic foot problems and dealing with foot problems that do not need to be treated in hospital.

The foot protection service should be led by a podiatrist (someone trained to look after your feet; sometimes called a chiropodist) with special training in dealing with diabetic foot problems.

### Multidisciplinary foot care service

The multidisciplinary foot care service is usually based at a hospital. It specialises in treating more severe diabetic foot problems. You might also hear it called an interdisciplinary foot care service, but the meaning is the same.

The terms 'multidisciplinary' and 'interdisciplinary' mean that the foot care service is made up of different kinds of healthcare professionals who work together to deliver care. The diabetes multidisciplinary service should include diabetologists (consultants who specialise in diabetes), podiatrists and diabetes nurse specialists, as well as other healthcare professionals with expertise in treating and managing diabetic foot problems.

Some treatments or care described here may not be suitable for you. If you think that your treatment does not match this advice, talk to your healthcare professional.

#### Questions to ask about your care

- Please tell me more about living with diabetes.
- Please tell me more about diabetic foot problems.
- Can you give me written information? What websites are helpful?
- What about lifestyle? Can you give me information about diet and exercise?
- How often will I need to see someone? Who will I see and where will I see them?
- What should I take to my appointments?
- Who should I contact if I have any questions or am worried about my diabetes or about my feet?
- What sorts of health problems are associated with having diabetes? What can I do to help prevent these?
- Can you point me towards local or national support groups?

## Foot checks for people with diabetes

If you have diabetes, it's very important to have your feet and lower legs checked regularly, to make sure there are not any problems.

### How often should foot checks happen?

Adults with diabetes should have a foot check:

- when diabetes is diagnosed and at least once a year after that
- if they think they have a problem with their feet
- if they have to go into hospital for any reason, and if they have any foot problems during the hospital stay.

Young people with diabetes who are aged 12 to 17 years will be looked after by the

hospital paediatric team (which looks after children and young people) or the hospital transitional care team (which looks after young adults who are preparing to move to adult services). They should have a foot check once a year as part of their diabetes annual review, and be given information about foot care. If they have any problems with their feet they should be referred to see a specialist.

Children under 12 (and their parents and carers) should be given information about foot care.

### What does the foot check involve?

The foot checks will usually be done by the foot protection service.

You'll need to take off your shoes and socks, and any bandages or dressings will be removed. Then, both your feet should be carefully examined. This will involve:

- finding out whether you have any foot problems at present
- examining your foot shape and footwear to see whether you may be at risk of rubbing or pressure
- checking your skin for changes in colour and looking for ulcers, sores, areas of hard skin and any signs of inflammation or infection
- testing the feeling in your feet to see how well the nerves are working
- taking the pulse in each of your feet to check the blood flow
- working out your risk (low, moderate or high) of developing a diabetic foot problem.

If the foot check shows that you do not have any foot problems, you will still need to have a foot check every year. Your healthcare professional should talk with you about your risk of developing foot problems in the future and explain how to look after your feet.

If you do have a foot problem, or if the foot check shows that you have a moderate or high risk of a problem developing, you may be referred to see another healthcare professional. This may be someone in the foot protection service or the multidisciplinary foot care service.

If you do not need to be referred to see anyone else, your healthcare professional will let

you know when they need to see you again.

### Importance of foot care

It is very important to take good care of your feet, because having diabetes puts you at risk of foot problems.

You should get information on diabetes and foot care when diabetes is diagnosed, during foot checks and if you develop any foot problems. The information should include:

- why foot care is important
- what good foot care involves
- what to do if you think there may be a problem
- footwear advice
- your risk of developing a foot problem in the future
- information about diabetes and the importance of managing your blood glucose.

The information you need about foot care – and how often you need to have a foot check – will depend on your risk of developing foot problems. Your healthcare professional can give you more information about this and talk with you about questions or worries you might have.

If you already have a foot problem (for example, an ulcer on your foot), your healthcare professional should explain what is wrong and what the treatment will involve. They may also ask whether you would find it helpful to see pictures of foot problems, such as long-term foot damage or amputation.

#### Questions to ask about your risk of developing a foot problem

- What is my risk of developing a foot problem, and how have you worked this out?
- What did the foot check show?
- How will I know if my risk changes?
- What should I do if I think I've got a foot problem?
- What is the best way of looking after my feet?
- Can you advise me about shoes, socks and insoles?
- Are there any changes to my lifestyle that I need to make?
- How can managing my diabetes help?
- How often do I need to have a foot check?
- Why are you referring me to someone else? What will they do?

### Treating diabetic foot problems

If you have a diabetic foot problem, you will be treated either by the foot protection service or the multidisciplinary foot care service.

Many foot problems can be treated by the foot protection service. However, if the diabetic foot problem is severe or if you have other problems that are a concern, then the multidisciplinary foot care service provides the treatment, usually in the hospital.

You should be referred to hospital immediately, and the multidisciplinary foot care service should be informed, if:

• You have a diabetic foot ulcer and you have a fever or any other symptoms of blood poisoning (the medical name for this is sepsis), or if there is a problem with the blood supply to your foot.

• Your healthcare professional thinks you have a severe foot or bone infection, or you have gangrene (whether or not you have a diabetic foot ulcer).

#### Questions to ask about treating diabetic foot problems

- Can you tell me more about the foot problem I have?
- How can making sure my diabetes is controlled help?
- What can I do to keep my diabetes under control?
- Can you tell me more about the treatment? What will it involve?
- What are the pros and cons of having this treatment?
- How long will the treatment take to have an effect?
- Are there any risks associated with this treatment?
- Are there any support organisations in my local area?
- Can you provide any information for my family and carers?

# **Diabetic foot ulcer**

If you have a diabetic foot ulcer, your healthcare professional should check the size and depth of the ulcer and look for signs of infection and other problems.

Treatment for diabetic foot ulcers involves 1 or more of the following:

- offloading, in which a plaster cast is used to take pressure off the ulcer and help with healing
- treating any foot infection with antibiotics
- making sure the foot has a good blood supply
- clearing away the dead, damaged and infected skin from the ulcer (the medical name) for this is debridement)

of 11

• using dressings while the ulcer heals.

The treatment will depend on how severe the ulcer is, where it is, and what you would prefer.

You'll need to have regular appointments as part of your treatment plan – how often will depend on your overall health, how well the ulcer heals, and whether any other problems develop.

## **Diabetic foot infection**

If your healthcare professional thinks you have a foot infection you should start antibiotics as soon as possible. The type of antibiotic will depend on how bad the infection is and whether there is a risk of complications.

A small sample should be taken from your foot to check which bacteria are causing the infection. Your doctor may change your antibiotic when the results of these tests come back.

The infection should start to clear up within 1 to 2 days of starting antibiotics. Tell your doctor if it does not start to improve, if it starts to get worse or if you feel very unwell at any time.

How long you have to take antibiotics will depend on how bad the infection is and how quickly it starts to improve. Your doctor should explain that antibiotics can cause side effects, such as diarrhoea and nausea (feeling sick).

You might also have an X-ray to see how severe the foot infection is. Some people may need a type of scan called an MRI.

# Charcot arthropathy

Charcot arthropathy is a serious condition, which is more common if people lose feeling in their feet. The bones in the foot can become weak and lead to dislocations, fractures and changes in the shape of the foot or ankle. Charcot arthropathy may develop if you have diabetes and you fracture your foot or ankle. If it is not treated quickly, ulcers or other sores can develop. Symptoms of Charcot arthropathy include the foot feeling hot and painful, and looking swollen or red.

Charcot arthropathy is diagnosed and treated in hospital by the multidisciplinary foot care service. If a healthcare professional thinks you may have Charcot arthropathy, they should refer you to the multidisciplinary foot care service within 1 working day, and you should be seen within another working day. You should rest and not put any weight at all on the foot until your appointment with the multidisciplinary foot care service. You may have an X-ray or a type of scan called an MRI.

The treatment for Charcot arthropathy usually involves having a plaster cast fitted.

# If you are treated in hospital

You should have a named consultant who will be responsible for your overall care.

If a diabetic foot problem is the reason you are in hospital, or if you have other medical problems but your diabetic foot problem needs the most urgent care, your care should be transferred to the multidisciplinary foot care service within 24 hours.

# Sources of advice and support

<u>Diabetes UK has more information about taking care of your feet</u>. You can also go to the <u>NHS website</u> for more information. The organisations below can give you more advice and support.

- Diabetes UK, 0345 123 2399
- Diabetes Research and Wellness Foundation

To share an experience of care you have received, contact your local <u>Healthwatch</u>.

NICE is not responsible for the content of these websites.

# Update information

**January 2023:** We created a new introduction section on diabetic foot problems: the care you should expect.

**November 2019:** We updated the information on antimicrobial prescribing for adults with a diabetic foot infection in line with the October 2019 update to the guideline.

ISBN: 978-1-4731-4946-5

### Accreditation

